Author’s response to reviews

Title: Recurrence in Unicentric castleman's disease postoperatively: a case report and literature review

Authors:

Na Ren (renna8581469@163.com)
Lei Ding (zr_dingl@163.com)
Erna Jia (jiaerna1012@163.com)
Jinru Xue (jinruxue@jlu.edu.cn)

Version: 2 Date: 22 Dec 2017

Author’s response to reviews:

dear editor and reviewer Umberto Bracal:

1 we have provided email addresses for all the authors on the title page.

2 we have changed 'Introduction' to 'Background' in the main text.

3 we have revised the two sections 'Discussion and Conclusions' into one section in the main text.

4 we have added the list of Abbreviations before the declarations section.

5 we have send "Consent for publication" to the email address of lingling.tian@biomedcentral.com

6 we have revised authors' names using their full initials in the Authors’ Contributions section.

7 we have revised the figures and figure legends as requested.